Author's response to reviews

Title: A novel copper complex induces ROS generation in doxorubicin resistant Ehrlich ascitis carcinoma cells and increases activity of antioxidant enzymes in vital organs.

Authors:

Ananda Mookerjee Mookerjee (soumitra01@yahoo.com)
Jayati Basu (soumitra01@yahoo.com)
Surajit Majumder (soumitra01@yahoo.com)
Shilpak Chatterjee (soumitra01@vsnl.net)
Gourisankar S Panda (soumitra01@vsnl.net)
Pranabananda Dutta (soumitra01@vsnl.net)
Smarajit Paul (smarajit@hotmail.com)
Pratima Mukherjee (pratimamukherjee@rediffmail.com)
Thomas Efferth (t.efferth@dkfz-heidelberg.de)
Shyamal Roy (srroy@iicb.edu.com)
Soumitra K Choudhuri (soumitra01@vsnl.net)

Version: 7 Date: 23 October 2006

Author's response to reviews: see over
To
The Editor, BMC Cancer 23.10.2006

Sir,
Thank You for accepting our paper in your prestigious journal. I am sending herewith the revised paper.

Looking forward to hearing from you,

Best regards
Soumitra Choudhuri
CNCl, Calcutta.